Author's response to reviews

Title: Psychometric properties of two physical activity questionnaires, the AQuAA and the PASE, in cancer patients

Authors:

- Roberto DK Liu (liurob@hotmail.com)
- Laurien M Buffart (l.buffart@vumc.nl)
- Marjolein Spiering (m.spiering@amc.uva.nl)
- Marie Jose Kersten (m.j.kersten@amc.uva.nl)
- Johannes Brug (j.brug@vumc.nl)
- Willem van Mechelen (w.vanmechelen@vumc.nl)
- Mai JM Chinapaw (m.chinapaw@vumc.nl)

Version: 6 Date: 11 March 2011

Author's response to reviews: see over
Dear editor,

We thank you for allowing us to make a final check of our manuscript entitled “Psychometric properties of two physical activity questionnaires, the AQuAA and the PASE, in cancer patients”.

A detailed overview of the typographical changes we have made to the revised manuscript is given below.

Kind regards,
On behalf of all co-authors,
Laurien Buffart,
Corresponding author.

Corresponding author’s contact information:
Laurien M Buffart,
EMGO Institute for Health and Care Research,
VU University Medical Center, Department of Epidemiology and Biostatistics
P.O. Box 7057, 1007 MB Amsterdam, The Netherlands.
E-mail: l.buffart@vumc.nl, Phone: +31 (0)20 444 9931, Fax: +31 (0)20 444 8181.
On page 2 under the subheading “Results”; in the 3rd sentence an “=” was placed between $r_s$ and 0.05.

On page 4 under the subheading “Study sample”; in the first sentence “haematology, oncology and gynaecology” were replaced by “Hematology, Oncology, Radiotherapy and Gynaecology”.

On page 6 under the subheading “Activity Questionnaire for Adults and Adolescents (AQuAA)”; in the last sentence “rs” was changed into “$r_s$”.

On page 11 under the subheading “Construct Validity”; in the 5th sentence “ICC=0.00” was changed into “ICC=-0.001” and in the 6th sentence “$r_s=0.16$, $p=0.279$ and ICCs= 0.12 and 0.39” was added.

In Table 2; “SDD95 and SDD95/range95 ratio” were changed into “SDD$_{95}$ and SDD$_{95}$/range$_{95}$ ratio”. The legend was also changed accordingly.